Author's response to reviews

Title: Oral maxillofacial neoplasms in an East African population a 10 year retrospective study of 1863 cases using histopathological reports.

Authors:

Adriane Kamulegeya (adrianek55@163.com)
Boniphace Kalyanyama (kalyanyamab@live.com)

Version: 10 Date: 7 July 2008

Author's response to reviews:

just made a few corrections in spellings.